Author's response to reviews

Title: Objective Structured Clinical Examinations (OSCEs), Psychiatry and the Clinical Assessment of Skills and Competencies (CASC). Same evidence, different judgement

Authors:
Steven Marwaha (s.marwaha@warwick.ac.uk)

Version: 3 Date: 11 May 2011

Author's response to reviews:

Tel: 024 7657 4885
Dr Steven Marwaha
Associate Clinical Professor of Psychiatry
Rm 149, Medical School Building
Gibbet Hill Campus
Warwick Medical School
Coventry
CV47AL
11th May 2011

Editor
BMC Psychiatry

Dear Editor,

Re: OSCEs, Psychiatry and the CASC. Same evidence, different judgement

Thank you for your email of the 10th May 2011 asking for formatting changes to made to the above manuscript.

In response to these:

1. I have now made the discussion and summary headings clear in the abstract.
2. I have removed highlighting and tracking from the manuscript.
3. I have now made a final check for typographical errors and made any corrections or minor amendments in language.

Yours sincerely

Dr Steven Marwaha